Syllabus: 22M:034 Engineering Math IV: Differential Equations

Fall 2003 (Section 102) TR 10:55 - 12:10, 118 MLH

Office: Phone: Email: idarcy@math.uiowa.edu Tentative Office Hours:

Course WWW site:

copies of course handouts.

Text edition (2001).

Grading system:

HW & Quizzes: 13% Computer Projects: 13% 80% <= B- < 81% <= B < 89% <= B+ < 90% 70% <= C- < 71% <= C < 79% <= C+ < 80% 60% <= D- < 61% <= D < 69% <= D+ < 70% F < 60%

The final is tentatively set for Schedule of Courses however, when the firm Final Exam Schedule

within one week from when the exam, homework, etc. has been handed back to the class (whether or exam, homework, etc).

More on computer projects

Maple

and/or computer algebra package (e.g. Mathematica

Sections to be covered

Attendance and absences : expected. announcements made during class;

secretary in 14C MLH.

acknowledged.

Policy on Scholastic Dishonesty:

Assignment 1:

Read Sections 1.1 - 1.4; p.8 #1,2,11,14,21,22; p.14 #1,6,7,8,12,13; and TBA